Point-by-point responses

Manuscript Number: 75649

Title: Cumulative incidence and risk factors for pouch adenomas associated with familial adenomatous polyposis following restorative proctocolectomy

Ryu et al. Incidence and risk factors for pouch adenoma

Dear Editor-in-Chief Andrzej S Tarnawski

We resubmit the revised manuscript, “Cumulative incidence and risk factors for pouch adenomas associated with familial adenomatous polyposis following restorative proctocolectomy,” to be considered for publication in World Journal of Gastroenterology. Corrections and other changes in the revised manuscript are indicated with red text. This letter contains point-by-point responses with corrections and additions.

Reviewer #1:

Q1. English needs revision in the whole manuscript.

A1. As per your suggestion, the whole manuscript has been edited by a professional language editor.

Q2. Please make a better sentence expressing the association of duodenal adenomas and pouch adenomas in the following sentence: "In addition, patients with duodenal adenomas developed pouch adenomas more frequently (66.7% vs 42.3%, p=0.039)." Assess if you tried to state that the presence of duodenal adenomas was associated with pouch adenomatosis?

A2. Thank you. We have rephrased the flagged passage. (Page 9, Line 9-11)

Original: patients with duodenal adenomas developed pouch adenomas more frequently (66.7% vs 42.3%, p=0.039).

Revised: patients with pouch adenomas (relative to those without pouch adenomas) were more likely to have duodenal adenomas (66.7% vs 42.3%, P = 0.039).

In patients with pouch adenomas, duodenal adenomas were more likely to be present. However, multivariate analysis showed no association between pouch adenoma and duodenal adenoma development. We have described this in the Discussion section. (Page 11, Line 13-16)
Reviewer #2:

Q1. Results. Paragraph: Clinical characteristics of the study patients with and without pouch adenomas. Line 9. In patients without pouch adenoma, the use of NSAIDs was not significantly more used for the treatment of desmoid tumors, according to the statistical criteria stated by the authors in the methods section of the manuscript, though the frequency is undoubtedly higher and near the statistical significance.

A1. Thank you for the comment. The sentence was incorrect, and we have corrected the description. (Page 9, Line 10-13)

Original: In patients without pouch adenomas, nonsteroidal anti-inflammatory drugs (NSAIDs) including celecoxib and meloxicam was significantly more used as a treatment for desmoid tumors (4.2% vs 19.7%, p=0.07).

Revised: Nonsteroidal anti-inflammatory drug (NSAID) treatment for desmoid tumors—including with celecoxib and meloxicam—was more common among patients without pouch adenomas, but this difference was not statistically significant (4.2% vs 19.7%, P = 0.07).

Q2. Fig. 1. The number of patients at risk is not well aligned with time after IPAA. This should be corrected.

A2. The number of patients is correct. We think that you are referring to the alignment of the box of excluded patients before IPAA, and we have edited Figure 1 accordingly. If our understanding is incorrect, please let us know, and we will make further modifications, as necessary.

Q3. Table 2. I do not understand what is the difference between: number of colorectal polyps, and colorectal polyps $<$ or $>$ 1000. According to the table in the group of patients without pouch adenoma the upper limit of the range is 500, whereas, below, 7 such patients had 1000 or more adenomas at the time of surgery. The authors should explain this discrepancy, or better define the two parameters.

A3. The number of colorectal polyps is expressed as an interquartile range (1st quartile-3rd quartile). Since 500 is the number corresponding to the 3rd quartile, not the upper limit, there were seven patients with more than 1000 polyps in the group of patients without pouch adenomas. The number of polyps was a variable that indicated the overall distribution. Also,
we considered that 1000 or more colorectal polyps at the time of surgery was clinically significant. Therefore, it was included in the analysis as a dichotomous (≥1000 and <1000) independent variable.

Q4. English polishing and corrections are needed all along the manuscript.

A4. As per your suggestion, the entire manuscript has been edited by a professional language editor.

Reviewer #3:

Q1. What was the indication for RPC-IPAA for FAP? was it for adenoma prophylaxis, cancer finding or presence of >100 polyps or both. Can the authors clarify that?

A1. The indication for RPC/IPAA was a diagnosis of FAP with or without malignancy. As we mention in the Method sections, the FAP patients were identified by presence of more than 100 colorectal polyps (Page 5, Lines 10-11). Some of them had cancer. The indication for RPC-IPAA has been added to the Methods section (Page 5, Lines 11-12).

Q2. The presented data should be integrated in Cox multivariate analysis and the multivariate analysis reported in Table 3 is the appropriate to be used. • Why duodenal adenoma was entered in the multivariate (despite a p = 0.54 in the univariate). Why?

A2. As per your suggestion, we have included more information regarding which variables were analyzed in the multivariable model. Variables reported to be associated with pouch adenoma development in previous reports and demographic variables were adjusted in the multivariable analysis. If we considered variables to be of theoretical importance, we entered them into the model despite a lack of statistical insignificance. After backward elimination, colorectal polyps more than 1000 and duodenal adenoma remained. We have included this information in the Methods section.

Q3. A comment is needed regarding presence of small bowel polyps at videocapsule endoscopy correlated with severity of Spiegelman classification, this agrees with the finding in Table 2 about duodenal adenomas.

A3. Thank you for important comment. Unfortunately, video capsule endoscopy is not a routine procedure at our institution. Most patients were not assessed for small bowel polyps with video capsule endoscopy; therefore, it would be inappropriate for us to comment on this matter.
Q4. Clinical characteristics of the study patients according to presence of pouch adenomas, the statistics and figures need to be carefully revisited prior to publishing this paper.

A4. We have reviewed this information.

Q5. Minor language/grammatic issues.

A5. As per your suggestion, this manuscript has been edited by a professional language editor.

Reviewer #4:

Q1. A radical linguistic revision by an English native speaker is necessary.

A1. As per your suggestion, the entire manuscript has been edited by a professional language editor.

Q2. In the “study population” paragraph of Methods section, Authors have reported the process of patients selection. However, this is a result and it should be moved in the appropriate section. In the above mentioned paragraph, Authors must only report inclusion and exclusion criteria.

A2. Thank you. As per your suggestion, the patient selection process has been moved to the Results section.

Q3. Please report which was the indication to proctocolectomy: prophylaxis, cancer finding or presence of >100 polyps. This variable should be integrated in Cox multivariate analysis as well.

A3. The indication for RPC/IPAA was a diagnosis of FAP with or without malignancy. As we mention in the Method sections, the FAP patients were identified by presence of more than 100 colorectal polyps (Page 5, Lines 10-11). Some of them had cancer. The indication for RPC-IPAA has been added to the Methods section (Page 5, Lines 11-12).
As per your recommendation, the multivariate analysis was performed again and included the presence of malignancy. Malignancy was excluded from step 5 during backward elimination, and the final result of the analysis did not change. Although progression to malignancy may be associated with pouch adenoma development, previous publications do not report malignancy as a risk factor for pouch adenoma, and we did not modify the table because there was no change in results.

<table>
<thead>
<tr>
<th>Variable</th>
<th>B</th>
<th>Wald</th>
<th>P</th>
</tr>
</thead>
<tbody>
<tr>
<td>Age</td>
<td>-0.06</td>
<td>0.18</td>
<td>0.67</td>
</tr>
<tr>
<td>Sex</td>
<td>-0.02</td>
<td>1.10</td>
<td>0.29</td>
</tr>
<tr>
<td>Desmoid tumor</td>
<td>-0.82</td>
<td>3.05</td>
<td>0.05</td>
</tr>
<tr>
<td>Gastric polyps</td>
<td>2.23</td>
<td>1.68</td>
<td>0.14</td>
</tr>
<tr>
<td>Duodenal adenomas</td>
<td>7.44</td>
<td>2.73</td>
<td>0.05</td>
</tr>
<tr>
<td>FIU, mo</td>
<td>-0.04</td>
<td>0.86</td>
<td>0.36</td>
</tr>
<tr>
<td>Colorectal cancer</td>
<td>-0.03</td>
<td>0.75</td>
<td>0.39</td>
</tr>
</tbody>
</table>

**Q4.** I do not understand why, in the multivariate analysis reported in Table 3, duodenal adenoma was entered in the multivariate despite a p = 0.54 in the univariate. Authors should
enter multivariate analysis parameters significant or close to statistical significance at univariate, or variables with a relationship plausibility such as NSAID use.

**A4.** As per your suggestion, we have included more information regarding which variables were analyzed in the multivariable model. Variables reported to be associated with pouch adenoma development in previous reports and demographic variables were adjusted in the multivariable analysis. If we considered variables to be of theoretical importance, we entered them into the model despite a lack of statistical insignificance. After backward elimination, colorectal polyps more than 1000 and duodenal adenoma remained. We have included this information in the Methods section.

**Q5.** Considering that presence of small bowel polyps at videcapsule endoscopy correlated with severity of Spiegelman classification, this agrees with the finding in table 2 about duodenal adenomas. Please comment.

**A5.** Thank you for important comment. Unfortunately, video capsule endoscopy is not a routine procedure at our institution. Most patients were not assessed for small bowel polyps with video capsule endoscopy; therefore, it would be inappropriate for us to comment on this matter.